Author’s response to reviews

Title: An unusual presentation of multiple cavitated lung metastases from colon carcinoma.

Authors:

Patrizia Seminara (patrizia.seminara@uniroma1.it)
Gaia Manna (gaia.manna@hotmail.it)
Alessandra Emiliani (emiliani.alessandra@libero.it)
Alessandro Iannace (ale_iann@hotmail.com)
Tania Losanno (tania.los@libero.it)

Version: 4 Date: 19 May 2011

Author’s response to reviews:

In reply to your last revision of the case report “An unusual presentation of multiple cavitated lung metastases from colon adenocarcinoma”, we would point out that in the paragraph evidenced by reviewer (the second of authors’ discussion), we made no reference to cavitated metastases. We referred to the possibility that pulmonary metastases from gastrointestinal adenocarcinomas can spread into the lung along the intact alveolar walls (lepidic growth) with radiologic imaging of air-space nodules.

However, we agree with you that the reader might be confounded by it and we accept to cut this paragraph entirely because its elimination doesn’t prejudice the complete discussion of the case report.